Author’s response to reviews

Title: Sleep Duration, Vital Exhaustion and Perceived Stress Among Pregnant Migraineurs and Non-Migraineurs

Authors:

Michelle A Williams (mwilliam@u.washington.edu)
Sheena K Aurora (sheena.aurora@swedish.org)
Ihunnaya O Frederick (Ihunnaya.Frederick@swedish.org)
Chunfang Qiu (Chun-fang.Qiu@swedish.org)
Bizu Gelaye (bizu@u.washington.edu)
Swee May Cripe (smtang@u.washington.edu)

Version: 3 Date: 2 November 2010

Author’s response to reviews: see over
November 2, 2010

MS ID : 4652154337331260
Michelle A Williams, Sheena K Aurora, Ihunnaya O Frederick, Chunfang Qiu, Bizu Gelaye and Swee May Cripe
Title : Sleep Duration, Vital Exhaustion and Perceived Stress Among Pregnant Migraineurs and Non-Migraineurs

BMC-Series Editorial Production Team
BioMed Central

Dear BMC Editorial Production Team

Thank you for your attention to our abovementioned manuscript and e-mail of November 2, 2010. We have revised our manuscript in light of your requests for formatting changes.

Sincerely,

Michelle A. Williams, ScD
Professor of Epidemiology and Global Health
Director, Multidisciplinary International Research Training Program
Director, Reproductive, Perinatal and Pediatric Training Program
Co-Director, Center for Perinatal Studies, Swedish Medical Center